Author's response to reviews

Title: Retention of Healthcare Professionals in Urban India Using Health Information Technology - A Path Analysis Study

Authors:

Indrajit Bhattacharya (indrajitbhat@gmail.com)
Anandhi Ramachandran (anandhirama@gmail.com)
B.K. Jha (bkiha44@gmail.com)

Version: 8 Date: 28 June 2015

Author's response to reviews: see over
We like to thank the reviewer for making efforts and investing time in our paper and providing valuable suggestions for improvement.

**Reviewer:** Marie-Pierre Gagnon
We have addressed all the concerns raised and suggestions made. We have rearranged the paragraphs, the lines starting with numbers, restricted use of the word and checked for capitals and hyphens. References to other studies have been moved from conclusion to discussion. The recommendations have been reconstructed as potential strategies that could be adopted. We have had our paper checked by a English editor.

I hope these would answer the reviewer’s comments.

**Reviewer:** Josue Mbonigaba

The reviewer’s main concern was our English. We have addressed this by seeking professional help from an English editor and a senior research scientist with more than 30 publications to his forte.